Author's response to reviews

Title: The Management of Large Perforations of Duodenal Ulcers

Authors:
Sanjay Gupta (sandiv99@yahoo.co.uk)
Robin Kaushik (robinkaushik@yahoo.com)
Rajeev Sharma (rjvsharma@yahoo.com)
Ashok KUMAR Attri (akattri@indmedica.com)

Version: 2 Date: 11 March 2005

Author's response to reviews:

The manuscript has been revised and rewritten to include the comments made by the referees. The references have also been updated and renumbered to correspond with the text.